Author's response to reviews

Title: The effects of alcohol consumption, psychological distress and smoking status on emergency department presentations in New South Wales, Australia

Authors:

Devon Indig (d.indig@med.unsw.edu.au)
Margo Eyeson-Annan (meyes@doh.health.nsw.gov.au)
Jan Copeland (j.copeland@unsw.edu.au)
Katherine M Conigrave (katec@med.usyd.edu.au)

Version: 3 Date: 25 March 2007

Author's response to reviews: see over
25 March 2007

Editorial Director, BMC Public Health
Professor Deborah Saltman
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Professor Saltman,

Thank you for your decision to publish our manuscript titled ‘The effects of alcohol consumption, psychological distress and smoking status on emergency department presentations in New South Wales, Australia’ in BMC Public Health.

I am writing to confirm that the requested formatting changes have been made to the manuscript. The requested changes include:

- Removal of ‘*These authors contributed equally to this work’ from the title page.
- Replacement of any underlining with bold in the manuscript
- Removal of table placement text.

We appreciate your efforts in reviewing our work and look forward its publication. Please direct all communication to the first author.

Sincerely,

Devon Indig
Phd Candidate